\T\begin{sidewaystable}
\T\scriptsize
\T\begin{threeparttable} 
\W\xmlattributes*{table}{class=myTable}

\begin{tabular}{lp{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{\maptoolspalte}p{1.2cm}} 

\T\toprule 
\label{tab:maptools}
&\multicolumn{2}{c}{Free \&}
&\multicolumn{9}{c}{Free \&}
&\texonly{\tiny}{proprietary\&}\\
&\multicolumn{2}{c}{100\% client-side}
&\multicolumn{9}{c}{client-serverside}
&\texonly{\tiny}{100\%client.}\\
\T\cmidrule(lr){2-3}\cmidrule(lr){4-12}\cmidrule(l){13-13} 

&\T\textbf{\begin{turn}{90}OpenLayers\end{turn}} \tiny{S.\pageref{maptool:openlayers}f}
\W\textbf{\link{OpenLayers}{maptool:openlayers}} 

&\T\textbf{\begin{turn}{90}WMS\end{turn}}\textbf{\begin{turn}{90}Mapper\end{turn}} \tiny{S.\pageref{maptool:wmsmapper}f}
\W\textbf{\link{WMS Mapper}{maptool:wmsmapper}}

&\T\textbf{\begin{turn}{90}CartoWeb\end{turn}} \tiny{S.\pageref{maptool:cartoweb}f}
\W\textbf{\link{CartoWeb}{maptool:cartoweb}}

&\T\textbf{\begin{turn}{90}Chameleon\end{turn}} \tiny{S.\pageref{maptool:chameleon}f}
    \W\textbf{\link{Chameleon}{maptool:chameleon}}

&\T\textbf{\begin{turn}{90}iGeoPortal\end{turn}} \tiny{S.\pageref{maptool:igeoportal}f}
\W\textbf{\link{iGeoPortal}{maptool:igeoportal}}

&\T\textbf{\begin{turn}{90}ka-Map\end{turn}} \tiny{S.\pageref{maptool:kamap}f}
\W\textbf{\link{ka-Map}{maptool:kamap}}

&\T\textbf{\begin{turn}{90}Mapbender\end{turn}} \tiny{S.\pageref{maptool:mapbender}f}
\W\textbf{\link{Mapbender}{maptool:mapbender}}

&\T\textbf{\begin{turn}{90}Mapbuilder\end{turn}} \tiny{S.\pageref{maptool:mapbuilder}f}
    \W\textbf{\link{Mapbuilder}{maptool:mapbuilder}}

&\T\textbf{\begin{turn}{90}MapGuide\end{turn}}\textbf{\begin{turn}{90}Open Source\end{turn}} \tiny{\pageref{maptool:mapguide}f}
\W\textbf{\link{MapGuide Open Source}{maptool:mapguide}}

&\T\textbf{\begin{turn}{90}Mapping\end{turn}}\textbf{\begin{turn}{90}Widgets\end{turn}} \tiny{\pageref{maptool:mappingwidgets}f}
\W\textbf{\link{Mapping Widgets}{maptool:mappingwidgets}}

&\T\textbf{\begin{turn}{90}p.mapper\end{turn}} \tiny{S.\pageref{maptool:pmapper}f}
\W\textbf{\link{p.mapper}{maptool:pmapper}}

&\T\textbf{\begin{turn}{90}Google Maps\end{turn}} \tiny{S.\pageref{maptool:googlemaps}f}
    \W\textbf{\link{Google Maps}{maptool:googlemaps}}
\\
\T\cmidrule(lr){2-2}\cmidrule(lr){3-3}\cmidrule(lr){4-4}\cmidrule(lr){5-5}\cmidrule(lr){6-6}\cmidrule(lr){7-7}\cmidrule(lr){8-8}\cmidrule(lr){9-9}\cmidrule(lr){10-10}\cmidrule(lr){11-11}\cmidrule(lr){12-12}\cmidrule(l){13-13}

Version
&2.3
&0.03
&3.3.0
&2.4.1
&1.2.1
&1.0
&2.4.1
&1.0.1
&1.1.0
&0.3.1
&3.0.1
&2.79\\ 

last Update
&21.02.07
&k.A.
&31.08.06
&06.09.06
&15.09.05
&05.02.07
&23.03.07
&19.07.06
&09.12.06
&17.03.06
&30.12.06
&18.04.07\\

FS-License
&BSD
&AFL
&GNU GPL
&X11
&GNU GPL
&MIT
&GNU GPL
&GNU LGPL
&GNU LGPL
&GNU GPL
&GNU GPL
&-\\


\T\midrule
Revision Administration
&SVN
&-
&CVS
&CVS
&SVN
&CVS
&SVN
&SVN
&SVN
&SVN
&SVN
&-\\

Developer Mailing List (ML)
&\checkmark
&-
&\checkmark
&-
&\checkmark\tnote{3}
&\checkmark
&\checkmark
&\checkmark
&\checkmark
&-
&-
&-\\

\hspace{\maptooltab}Mails per Month\tnote{1}
&85
&
&16
&
&49
&11
&55
&120
&168
&
&
&-\\

\hspace{\maptooltab}Active Developers\tnote{2}
&72
&
&12
&
&101
&7
&46
&47
&42
&
&
&-\\


User Mailing List (ML)
&\checkmark
&-
&\checkmark
&\checkmark
&\checkmark\tnote{3}
&\checkmark
&\checkmark
&\checkmark
&\checkmark
&-
&\checkmark
&\checkmark\\

\hspace{\maptooltab}Mails per Month\tnote{1}
&197
&
&80
&51
&92
&96
&107
&65
&527
&
&82
&-\tnote{4}\\

\hspace{\maptooltab}Active Users\tnote{2}
&150
&
&86
&63
&106
&121
&98
&78
&297
&
&61
&-\tnote{4}\\


\T\midrule
Overview Map
&\checkmark
&-
&\checkmark
&\checkmark
&\checkmark
&\checkmark
&\checkmark
&-
&-
&-
&\checkmark
&\checkmark\\ 

Layer Overview
&\checkmark
&-
&\checkmark
&\checkmark
&-
&\checkmark
&\checkmark
&\checkmark
&\checkmark
&-
&\checkmark
&\checkmark\\ 

Legend
&-
&-
&\checkmark\tnote{5}
&\checkmark
&\checkmark
&\checkmark\tnote{5}
&\checkmark
&-
&\checkmark\tnote{5}
&\checkmark
&\checkmark
&-\\ 

Scale Bar
&\checkmark / -
&- / -
&\checkmark / \checkmark
&- / \checkmark
&- / -
&\checkmark / \checkmark
&\checkmark / \checkmark
&\checkmark / -
&\checkmark / -
&-
&\checkmark / \checkmark
&- / \checkmark\\ 




Zoom Bar
&\checkmark
&-
&-
&-
&-
&-
&-
&-
&\checkmark
&-
&\checkmark
&\checkmark\\ 

Pan Navigation\\ 
\hspace{\maptooltab}on map edge
&-
&-
&\checkmark
&\checkmark
&\checkmark
&-
&\checkmark
&-
&-
&-
&-
&-\\ 

\hspace{\maptooltab}in PanZoom Bar
&\checkmark
&-
&-
&-
&-
&-
&-
&-
&\checkmark
&-
&-
&\checkmark\\ 

\T\midrule
Zooming per\\ 
\hspace{\maptooltab}Double-click
&\checkmark
&-
&-
&-
&-
&-
&-
&-
&-
&-
&-
&\checkmark\\ 

\hspace{\maptooltab}Mousewheel
&\checkmark
&-
&-
&-
&-
&\checkmark
&-
&-
&-
&-
&\checkmark
&\checkmark\\ 

\hspace{\maptooltab}Zoom Box (Shift-key)
&\checkmark (\checkmark)
&-
&\checkmark (\checkmark)
&\checkmark (-)
&\checkmark (-)
&\checkmark (-)
&\checkmark (-)
&\checkmark (\checkmark)
&\checkmark (\checkmark)
&\checkmark (-)
&\checkmark (\checkmark)
&-\\ 

Panning with\\ 
\hspace{\maptooltab}Overview Map
&\checkmark
&-
&\checkmark
&\checkmark
&\checkmark
&\checkmark
&\checkmark
&-
&-
&-
&\checkmark
&\checkmark\\ 

Zooming \& Panning\\ \hspace{\maptooltab}with Keyboard
&\checkmark
&-
&-
&-
&-
&-
&-
&-
&-
&-
&\checkmark
&\checkmark\\ 


\T\midrule
Tiling
&\checkmark
&\checkmark
&-
&-
&-
&\checkmark
&-
&-
&\checkmark
&-
&-
&\checkmark\T\\ 
\T\bottomrule 
\texorhtml{\caption}{\htmlcaption}{Listing of Analysis Results of Web Mapping Applications}

\end{tabular} 
\T\begin{tablenotes}\tiny
    \W\begin{menu}  \small  
    \item[1] Average monthly mail volume Oct 2006 - Mar 2007
         (6 months)\hspace{9.5cm} As of: \checkDate
    \item[2] Total number of active developers/users in analyzed time period 
         (Oct 2006 - Mar 2007; 6 months)
    \item[3] Uses general user/developer ML of \textit{deegree}
    \item[4] Automatic analysis not possible in \textit{Google Groups}
    \item[5] Legend is integrated into layer overview
        \W\end{menu}
\T\end{tablenotes}    

\T\end{threeparttable}
\T\end{sidewaystable}

\T\normalsize
